Author's response to reviews

Title: Prevalence of tuberculosis and treatment outcome among university students in Northwest Ethiopia: A retrospective study

Authors:

Beyene Moges (beyemoges@gmail.com)
Bemnet A Tedla (amarebem6@gmail.com)
Gizachew Yismaw (gizachewyismaw@yahoo.com)
Meseret Workineh (mwmesil@gmail.com)
Shitaye Alemu (shitayea@yahoo.com)
Desalew Mekonnen (desalewm@yahoo.com)
Ermias D Ejjara (ermi_diro@yahoo.com)
Belay Tesema (bt1488@yahoo.com)
Afework K Gizaw (afeworkkassu@yahoo.com)

Version: 5 Date: 19 December 2014

Author’s response to reviews:

The ‘Ethical considerations’ section of the manuscript is modified as follows: ‘The study was conducted after ethical approval was obtained from Institutional Review Board of the University of Gondar. Permission of access to participants’ data was also obtained from the University of Gondar Hospital. No patient identifiers were used during data collection.’